Author’s response to reviews

Title: Severe Viperidae envenomation complicated by shock, acute kidney injury, and gangrene presenting late at the emergency department: a case report

Authors:

Agnès Esiéné (eseineagnes@gmail.com)
Paul Owono Etoundi (owanop@yahoo.com)
Junette Arlette Mbengono Metogo (junetmell@yahoo.com)
joel tochie (joeltochie@gmail.com)
Jacqueline Ze Minkande (minkandeze@yahoo.com)

Version: 2 Date: 22 Feb 2019

Author’s response to reviews:

22th February, 2019.

The Editor-in-chief,

BMC Emergency Medicine.

Dear Sir/Madam,

We are writing to submit our manuscript entitled “Severe Viperidae envenomation complicated by shock, acute kidney injury, and gangrene presenting late at the emergency department: a case report” for consideration for publication in BMC Emergency Medicine.

We have addressed all reviewers’ queries and have used the track changes options of MS Word to modify the paper so as to account for their comments and suggestions.

We hope that you will find this revised version suitable for publication.

Sincerely,

On behalf of all co-authors
Joel Noutakdie Tochie, MD.

Faculty of Medicine and Biomedical Sciences,

University of Yaoundé 1, Yaoundé, Cameroon.

1) Reviewer comments: add the figure legend after the reference section

Authors’ revision: page 14 line 13 &à 17: the Legend of Figures and Tables has been included